Author's response to reviews

Title: Prevalence of the metabolic syndrome among children from six cities of China

Authors:

Haiquan Xu (xhaiguan@163.com)
Yanping Li (liyanping72@yahoo.com)
Ailing Liu (liuailing72@yahoo.com)
Qian Zhang (zhangq-99@263.net)
Xiaoqi Hu (huxiaqee@163.com)
Hongyun Fang (fanghy1982@126.com)
Tingyu Li (tyli@vip.sina.com)
Hongwei Guo (hwguo@shmu.edu.cn)
Ying Li (liying_helen@163.com)
Guifa Xu (vyxu@sdu.edu.cn)
Jun Ma (majunt@bjmu.edu.cn)
Lin Du (dulinaa@hotmail.com)
Guansheng Ma (mags@chinacdc.cn)

Version: 6 Date: 31 December 2011

Author’s response to reviews:

Dear editor,

We have made some changes in the latest revised version as follows, thank you.

1. “type 2 Diabetes mellitus” was changed into “type 2 diabetes” in the Background part (Page 3, line 13).

2. “diferent” was changed into “different” for the title of Table 1 in the Results part. (Page 4, line 8, Table 1)

3. “Mean ± SD” was added in the Table 2 for the Total column in the Results part. (Page 6, line 22, Table 2)

4. “P” was changed into “p” in the note of Table 3 (Page 10, line 11) and Table 4 (Page 13, line 9) of the Results part.

5. “High Glucose level” was changed into “High glucose level” in the Table 3. (Page 8, line 2)

6. “YL1” and “YL4” were changed into “YLAff1” and “YLAff4”, respectively, in the Authors’ contributions part. (Page 15, line 31-33)

7. “Supplement Table 1” was changed into “Table S1” in the additional file 1.

Best wishes.
Happy New Year.
Guansheng Ma